International Journal of Linguistics, Literature, and Culture



## **REVIEW HISTORY**

Paper: "The Depiction of Native American in the Novels of Forrest Carter"

Corresponding Author: Suryo Sudiro Email: <a href="mailto:sudiro71@gmail.com">sudiro71@gmail.com</a>

Doi: 10.19044/llc.v7no1a2

Peer review: Reviewer 1: Tamari Dolidze Grigol Robakidze University, Georgia

**Reviewer 2: Blinded** 

Published: 31.03.2020

